Author's response to reviews

Title: Investigating the complex genetic architecture of ankle-brachial index, a measure of peripheral arterial disease, in non-Hispanic whites

Authors:

Sharon LR Kardia (skardia@umich.edu)
M. Todd Greene (mtgreene@umich.edu)
Eric Boerwinkle (Eric.Boerwinkle@uth.tmc.edu)
Stephen T Turner (sturner@mayo.edu)
Iftikhar J Kullo (kullo.iftikhar@mayo.edu)

Version: 4 Date: 5 May 2008

Author's response to reviews: see over
May 5, 2008

Dear Editor-in-Chief,

Please find enclosed our revised manuscript titled ‘Investigating the complex genetic architecture of ankle-brachial index, a measure of peripheral arterial disease, in non-Hispanic whites’ which we would like to re-submit as our final proofed version for publication in *BMC Medical Genomics* as an original research article. As part of our re-submission process, we have gone through the editorial comments and have made formatting changes as requested.

All authors have read and approved re-submission of the manuscript and the manuscript has not been published nor is not being considered for publication elsewhere.

None of the authors have any conflict of interest to declare.

Thank you very much for your consideration of the manuscript.

Sincerely,

Sharon L. Kardia, PhD